

Received 4 July 2024, accepted 13 July 2024, date of publication 17 July 2024, date of current version 29 July 2024. Digital Object Identifier 10.1109/ACCESS.2024.3429623

SURVEY

Findings on Machine Learning for Identification of Archaeological Ceramics: A Systematic Literature Review

ZIYAO LING, GIOVANNI DELNEVO[®], (Member, IEEE), PAOLA SALOMONI, AND SILVIA MIRPI[®] (Member IEEE)

AND SILVIA MIRRI[®], (Member, IEEE) Department of Computer Science and Engineering, University of Bologna, 40127 Bologna, Italy Corresponding author: Giovanni Delnevo (giovanni.delnevo2@unibo.it)

ABSTRACT The identification of archaeological ceramics is a relevant topic in the field of cultural heritage, and the history of archaeological ceramics can be traced back to prehistoric times. At present, there are two main methods for identifying archaeological ceramics, the empirical method and the technical one. In practice, these methods are costly or time-consuming. A systematic literature review of thirty-three studies on the identification of archaeological ceramics using machine learning is presented in this paper, including the collection process to build the dataset, the image processing of archaeological ceramic images, and the machine learning algorithms used for the classification. The main findings show the efficacy of deep learning for the automatic classification of archaeological ceramics compared to other approaches and highlight the need for more comprehensive and standardised datasets to further improve the automatic classification process.

INDEX TERMS Archaeological ceramic identification, archaeological ceramic classification, machine learning, deep learning, archaeological ceramics dataset.

I. INTRODUCTION

Ceramics refer to a broad category of materials, typically inorganic and non-metallic, employed by humans for over 10,000 years [1]. It includes pottery, majolica, faience, terracotta, stone mass, and porcelain [2]. Their remarkable durability has allowed them to withstand the test of centuries, preserving a window into the past and safeguarding the stories of our ancestors [3]. Consequently, ceramics serve as chronological markers that are indispensable for reconstructing the past and comprehending the temporal progression of cultures, from their emergence to their decline, thereby enriching our understanding of the human journey through time [4]. Moreover, since they were among the earliest commodities traded between different regions and cultures, they serve as invaluable artefacts that unveil intricate trade networks and interactions between ancient societies [5]. They trace the paths of cultural exchange, uncover the complexities of

The associate editor coordinating the review of this manuscript and approving it for publication was Mostafa M. Fouda^(D).

ancient trade routes, and highlight the interconnections of diverse civilizations across time and space [3]. For these reasons, it becomes of fundamental importance to accurately date and classify ancient ceramics.

Currently, the methods for the identification of archaeological ceramics, mainly focused on pottery and porcelain, can be grouped into two main families [6]. On the one hand, there is empirical identification [7]. It is a skilled process relying on the expertise of human analysts, that identify archaeological ceramics manually. Hence, it is a time-consuming activity, that often lacks objective constraints leading to inconsistent classifications. On the other hand, there are scientific identification methods that encompass techniques such as X-ray fluorescence analysis, thermoluminescence dating, and spectral analysis [8]. Thermoluminescence dating may cause damage to the porcelains themselves, resulting in irreversible damage [9]. While X-ray fluorescence analysis is a nondestructive technique [10], it does have some drawbacks for definitive identification: i) it primarily detects elements present in the porcelain and it misses subtle details, including

firing temperature, specific minerals used, or manufacturing techniques; ii) it hinges on the quality and completeness of the reference database used about the elemental makeup of known porcelains; iii) it typically analyzes the surface composition that can be altered because of weathering, restoration, or glazing with a different composition. Moreover, it requires expensive equipment, involves complex identification processes, and is time-consuming.

Hence, the identification of archaeological ceramics, including pottery and porcelain, faces two key challenges: it can be time-consuming and it heavily relies on expert knowledge and experience. This has led researchers to explore Computer Vision (CV) and Artificial Intelligence (AI) as potential tools to aid in the identification process. Thanks to the advancements in these fields [11], [12], [13], more studies investigated their use for the identification of archaeological ceramic images in recent years [14], [15], [16]. Just to cite a few, Wang et al. [17] applied the Support Vector Machine (SVM) algorithm to design a prototype for feature extraction and classification of porcelain images. Instead, Zheng et al. [18] employed a Gaussian colour model and multi-scale filter bank to extract colour texture features.

This paper presents a systematic literature review on ML algorithms for archaeological ceramics identification. The review is based on scientific papers available in the most prominent digital libraries in the computer science field. Papers published between 2013 and 2023 were considered. Following the PRISMA methodology guidelines, 33 papers were finally analysed in terms of data sets, image processing methods, and ML algorithms used.

The rest of the paper has the following structure. Section II presents some literature reviews in the context of archaeological ceramics. Section III describes the research questions that motivated this study and the followed methodology. Section IV then analyses the results of the study and discusses the main findings. Finally, Section V summarises the study and presents some future works.

II. BACKGROUND AND RELATED WORK

The application of ML and DL techniques in the field of archaeological ceramics identification is an interesting area of research. While our review seeks to provide a comprehensive overview, it is important to acknowledge previous works that have explored similar themes. Notably, several surveys and studies have been conducted to examine the intersection of archaeological analysis and computational techniques. These existing reviews serve as foundational references, offering valuable insights and highlighting the progress made in the application of ML and DL methodologies to archaeological ceramics identification.

Cumbajin et al. [19] presented a systematic literature review on deep learning for industrial surface defect detection of several materials, including ceramics. Such a review considered only the use of convolutional neural networks. Our paper differentiates itself by focusing exclusively on the identification and classification of archaeological ceramics, employing a variety of ML and DL algorithms tailored to the unique characteristics of archaeological data. While the industrial review proposes a new taxonomy for surface defect detection, our work seeks to develop and refine methodologies specifically for archaeological ceramics.

Zhang et al. [20] proposed a review on the use of ML the discovery of high-entropy ceramics. High-entropy materials, known for their versatile platform and superior structural and functional properties, present a vast phase space that complicates efficient design through traditional methods. The review highlights how ML accelerates the discovery and optimization of high-entropy ceramics. The paper discusses the entire ML pipeline from data collection and feature engineering to model refinement and performance prediction improvement. In contrast, our review delves into the application of ML algorithms for archaeological ceramics, which, unlike high-entropy ceramics, involves the analysis of artifacts with historical and cultural significance.

Di Angelo et al. in [21] conducted a comprehensive review of automatic methods for analyzing archaeological pottery sherds, encompassing the state-of-the-art up to the end of 2021. They provided a critical analysis of the most significant advancements in pottery analysis, classification, and reconstruction from a 3D discrete manifold model. In contrast to this review, our paper do not focus on 3D models but mainly on images.

Ming et al. [22] provided a review of the application of virtual reality (VR) technology in the conservation and restoration of archaeological ceramics. However, this review does not cover the application of ML algorithms for the identification of archaeological ceramics, although the existing review on VR technology is not directly related to ML, it still can provide future research directions in the field of archaeological ceramics, such as integration of ML algorithms with VR for the identification of archaeological ceramics.

III. METHODS

This section presents the research questions that motivated this research, describes the PRISMA methodology and how the data were collected and processed.

A. RESEARCH QUESTIONS

The main research question that drove this study is:

RQ: What are the main findings derived from the application of machine learning for the identification of archaeological ceramics?

In particular, this work aims to answer the following questions to shed light on three main aspects:

- **RQ1**: Which are the main datasets employed in the identification of archaeological ceramics?
- **RQ2**: What are the findings about the image processing of archaeological ceramics for its identification?
- **RQ3**: What are the main ML algorithms used in the identification of archaeological ceramics?

RQ1 will mainly be centred around ceramic type (pottery and porcelain), country, period, pattern or type, sample count, and data source. RQ2 will focus on different image processing methods and their frequency and tendency. Finally, RQ3 will mainly analyze ML or Deep Learning (DL) algorithms, including their distribution and tendency. Specific analyses and discussions of these three research questions will be presented in Section IV.

B. PRISMA METHODOLOGY

Systematic reviews aim to identify all research addressing a specific and well-defined question to provide a balanced and unbiased literature summary [23]. A systematic review takes advantage of a rigorous and transparent methodology, often outlined in a protocol, to identify, evaluate, and synthesize relevant research studies. The search for relevant studies is often carried out across various databases and sources. Finally, the studies are evaluated and the main findings are synthesized to provide a comprehensive overview.

In this work, we took advantage PRISMA (Preferred Reporting Items for Systematic reviews and Meta-Analyses) methodology [24], [25]. PRISMA provides a framework for the conduct of systematic reviews and future analyses in the field of research. It ensures transparency, reproducibility and decrease of bias in the review process and including an organized and complete approach to the organisation and presenting of systematic reviews.

C. DATA COLLECTION AND PROCESSING

We checked published English-language articles or materials that are available on the web through noticeable digital libraries and databases in the field of computer science. They include ACM Digital Library, IEEE Xplore, Scopus, ScienceDirect, SpringerLink, and CNKI. In addition to them, Google Scholar was used to supplement the search to minimise the risk of exclusion.

The following search keywords (Boolean logic) were entered into the six databases: ("computer vision" OR "artificial intelligence" OR "machine learning" OR "deep learning") AND ("archaeological ceramic" OR "ancient ceramic" OR "porcelain") AND ("identification" OR "classification" OR "recognition"). The search options were title, abstract and keywords for the Scopus database.

We included any conference or journal paper concerning computer vision and machine learning for the identification of archaeological ceramics that have been published during the last ten years (i.e., between 2013 and 2023). To ensure both quality and accuracy, only peer-reviewed journal articles for which the full text is available have been included. The other inclusion and exclusion criteria are listed in Table 1. It is important to notice that ML algorithms and general computer vision approaches must be employed to identify or classify archaeological ceramics and not for other tasks such as reconstruction or restoration. Moreover, we are interested in finding research articles and not other literature reviews.

TABLE 1. Inclusion and exclusion criteria.

Inclusion Criteria	Exclusion Criteria
Focus on image processing techniques and feature extrac- tion approaches for archaeo- logical ceramics images	Focus on chemical elements or spectra of archaeological ceramics
Use computer vision and ma- chine learning approaches for identification or classification of archaeological ceramics Empirical studies	Use computer vision and ma- chine learning approaches for reconstruction or restoration of archaeological ceramics Literature reviews, commen-
Written in English Peer-review	taries or meta-analysis Written in other languages Pre-print

The papers were searched by following four steps: i) collection, ii) screening, iii) eligibility, and iv) information extraction. A summary of the papers selected and excluded at each step of the PRISMA review process is reported in Figure 1. Firstly, we collected all the papers that corresponded to the search criteria. Secondly, the dataset with the most results was Google Scholar with 997 papers. The duplicated records removed (i.e., 56 papers) were found in Scopus, ACM Digital Library, IEEE Explore, Science Direct, Springer Link, and CNKI, and were also found on Google Scholar. From the remaining 1814 papers, we excluded the following ones:

- 1032 papers because: i) the language of the paper is not English, and ii) the title indicates that the paper is not focused on the identification of archaeological ceramics using computer vision techniques.
- 721 papers because at the abstract level, they were not focused on archaeological ceramics but modern ceramics.
- 18 papers because they were not focused on identification and classification but on other different tasks.
- 10 papers because they were not focused on images of archaeological ceramics.

Finally, thirty-three papers meeting the standard were eventually retained (listed in Table 11 in Appendix), labelled ID1-ID33 sequentially.

D. CRITICAL ASSESSMENT

Besides the selection process described in the Subsection III-C, an assessment of the quality criteria was also performed. The results are presented in Table 2, which has one column for each research question (RQ1, RQ2, and RQ3) and as many rows as articles. For each article resulting from the selection process, we assessed whether they included an answer to the different research questions (number 1 in the table), or not (number 0 in the table).

Regarding the RQ1, which focuses on the dataset, most of the papers well described the data details of the study, while three studies, namely ID5 [26], ID10 [27], and ID17 [28] did not provide adequate information. With regard to the RQ2, out of thirty-three papers, only ID13 [29] did not evidently describe the image processing methods used. Regarding the



FIGURE 1. Literature selection process following the PRISMA methodology.

RQ3, five studies lacked a clear description of the ML algorithm for the identification of ceramics. In conclusion, twenty-six studies provide enough information to answer all the research questions while the minority of studies (i.e., five ones) obtain a score of two out of three, and two studies ID10 [27] and ID17 [28] obtain a score of one out of three.

E. EXTRACTION OF RELEVANT FIELDS

Finally, the last step of our methodology is to elicit and summarize relevant information from the selected articles. This was done manually by reading the full texts of the selected articles. The elicited information includes the answers to the research questions of the article. The complete list of fields elicited from the articles is provided in Table 3. Three fields that were manually compiled in response to the different research questions.

IV. FINDINGS AND DISCUSSION

This Section presents an initial quantitative analysis regarding the retrieved papers and then the answers to the three research questions are discussed in isolation, in the following subsections.

A. QUANTITATIVE ANALYSIS

In this Section, we provide a quantitative analysis of various attributes of the retrieved papers, including the countries of origin of the datasets used in these studies.

TABLE 2. Critical assessment results of the reviewed studies.

ID	RQ1	RQ2	RQ3	Total
ID1 [30]	1	1	1	3
ID2 [31]	1	1	1	3
ID3 [32]	1	1	1	3
ID4 [33]	1	1	1	3
ID5 [26]	0	1	1	2
ID6 [34]	1	1	1	3
ID7 [35]	1	1	0	2
ID8 [36]	1	1	1	3
ID9 [37]	1	1	1	3
ID10 [27]	0	1	0	1
ID11 [38]	1	1	1	3
ID12 [39]	1	1	1	3
ID13 [29]	1	0	1	2
ID14 [40]	1	1	1	3
ID15 [41]	1	1	1	3
ID16 [42]	1	1	1	3
ID17 [28]	0	1	0	1
ID18 [7]	1	1	1	3
ID19 [43]	1	1	1	3
ID20 [44]	1	1	1	3
ID21 [45]	1	1	0	2
ID22 [46]	1	1	1	3
ID23 [47]	1	1	1	3
ID24 [48]	1	1	1	3
ID25 [49]	1	1	1	3
ID26 [50]	1	1	1	3
ID27 [51]	1	1	1	3
ID28 [52]	1	1	0	2
ID29 [53]	1	1	1	3
ID30 [14]	1	1	1	3
ID31 [54]	1	1	1	3
ID32 [55]	1	1	1	3
ID33 [56]	1	1	1	3

TABLE 3. Information extracted from the papers included in the systematic literature review.

	Field	Description
1	ID	Unique identifier for the study
2	Title	Title of the paper
3	Authors	Authors of the paper
4	Year	Year of publication
5	Country	Country of the first author of the
6	Keywords	paper The keywords listed by the authors in the paper
7	Main results	Main results presented in the paper
8	Information about the quality analysis	Criteria presented in Section III-D
9	RQI	Datasets employed in the archae- ological ceramics identification for computer vision
10	RQ2	Findings related to the image pro- cessing of archaeological ceramics for identification
11	RQ3	ML algorithms used in the identifi- cation of archaeological ceramics

As mentioned before, the research process retrieved thirty-three relevant studies. These studies employed datasets from twelve different countries across five continents. Table 4 shows the distribution of datasets used in these studies, classified by the country in which the datasets were collected. The country presenting the most datasets was China (n = 16),

TABLE 4. Number of datasets from each article per country.

Country	Number	of
	Datasets	
China	16	
United States of America	5	
France	2	
Greece	2	
Egypt, Honduras, Italy, Japan, Korea, New	1	
Zealand, South Africa, Thailand		
Total	33	

TABLE 5. Information about the Journals.

Journal	Publisher
Across Space and Time	Amsterdam University
*	Press
Advances in Mathematical Physics	Hindawi Limited
Applied Artificial Intelligence	Taylor & Francis
Archaeology in New Zealand	New Zealand Archaeology
	Association
Ceramics International	Elsevier
Current Applied Science and Tech-	King Mongkut's
nology	Institute of Technology
	Ladkrabang
IEEE Access	IEEE
International Journal of Heritage in	SAGE Publications
the Digital Era	
Journal of Archaeological Science	Elsevier
Journal of Archaeological Science:	Elsevier
Reports	
Journal of Ceramics	CNKI
Journal on Computing and Cultural	ACM
Heritage	
Journal of Computer Applications	Ubiquity Press
in Archaeology	
Journal of the Chinese Ceramic So-	CNKI
ciety	
Research Journal of Applied Sci-	Maxwell Scientific Orga-
ences, Engineering and Technology	nization
Revista Argentina de Clinica Psico-	Fundacion Aigle
logica	
Soft Computing	Springer

followed by the United States of America (USA) (n = 5), and two datasets from France and Greece respectively.

Of the thirty-three papers, ten of them were published in proceedings of computer sciences conferences. The others were all published in journals. Table 5 lists the different journals and their publishers. In contrast to the conference papers, the journals that published such articles are not exclusively computer science journals but range from archaeology to cultural heritage computing.

B. DATASETS FOR ARCHAEOLOGICAL CERAMICS IDENTIFICATION

This Section answers the first research question **RQ1**, "Which are the main datasets employed in the identification of archaeological ceramics?". The answer includes six main aspects of the selected dataset: ceramic type (porcelain and pottery), country, period, pattern or type, sample count, and data source. Table 6 presents the dataset employed in each paper. As expected, the mass of them (30 out of 33 ones, that is 90.9%) clearly designated the dataset used. For the other

three papers (9.0%), ID5, ID10, and ID17, the label "N/A" is used, since they were mainly focused on the image processing methods [26], [27].

As already anticipated, in the field of archaeology, ceramics include pottery, majolica, faience, terracotta, stone mass, and porcelain [2]. In the selected studies, the datasets mainly contain two categories, pottery and porcelain, which are distinguished in Table 6 by the labels "Pottery" and "Porcelain". If the type of ceramic is not specified, we used the label "Ceramic". For the other characteristics (Period, Pattern Types, Sample Count, Data Source), they are set to N/A if in the original paper such information is not specified. More detailed information about the dataset is listed in Table 12 in the Appendix V.

1) TYPE, COUNTRY, AND PERIOD OF CERAMICS

These datasets mainly contain two different categories of ceramics, porcelain and pottery. They cover a wide range of periods (from the Neolithic period to the early 19th century). The datasets belong to 12 countries, including the United States of America, China, France, Greece, Egypt, Honduras, Italy, Japan, Korea, New Zealand, South Africa, and Thailand.

Porcelain is a special type of ceramic, that originated in China during the Tang dynasty (ca.1332 to 1043 aBP) [2]. Among thirty papers, sixteen papers selected Chinese porcelains as the dataset, ranging from Tang to Song (ca.990 to 671 aBP), Yuan (ca.679 to 582 aBP), Ming (ca.582 to 306 aBP), and Qing (ca.306 to 38 aBP) dynasty. Figure 4 shows the distribution of dynasties in the porcelain datasets. Two papers focused on only one dynasty for the identification of porcelains. ID1 [30] classified the porcelains from the Yaozhou kiln of the Song dynasty while ID21 [45] identified the washer-type porcelains from the Ming dynasty. Meanwhile, seven papers discussed two or more dynasties for the identification of porcelains. ID3 [32] discussed the porcelain ewers from the Tang, Song, and Yuan dynasties, Figure 2 shows the samples of dataset in ID3. ID4 [33] classified porcelains with four different patterns from Ming and Qing dynasties, ID8 [36] identified porcelain shards from Yuan, Ming, and Qing dynasties, Figure 3 shows the samples of dataset of ID8, and ID11 [38] classified two types of porcelain vases from Ming and Qing dynasties. Moreover, ID13 [29] identified different types of porcelain from Song, Yuan, Ming, and Qing dynasties, ID20 [44] classified porcelains with Banana leaf patterns from Yuan, Ming, and Qing dynasties, and ID28 discussed porcelain jars from Tang, Song, Yuan, and Ming dynasties.

Different dynasties have different styles, and of course, the style of the porcelain depends on the kiln, so the shape of the porcelain, the colour, and its pattern will have different characteristics depending on the style. The Ming dynasty is the most studied dynasty with seven papers focused on it. The reasons are two-fold. Firstly, such a dynasty has the largest number of preserved porcelain samples, and, secondly, it is the second flourishing period in the development of Chinese



FIGURE 2. Samples of typical chinese porcelain ewers in ID3 [32].



FIGURE 3. Samples of archaeological porcelain pieces in ID8 [36].

porcelain, which is valuable for research. Meanwhile, the Tang, Song, and Yuan dynasties had not been studied as much in comparison, which also provides a direction for future research, it can be expanded by focusing on the three less studied dynasties to broaden the scope of research. In addition, almost all of the selected papers focus on only one feature of porcelain, such as different types of porcelain from the same dynasty [38], or different patterns of porcelain from the same dynasty [34], or the study of the same type of porcelain from different dynasties [32]. No research integrates these features, such as different types of porcelain from different dynasties with different patterns, to expand the depth of the study of porcelain, which is also one of the directions that can be researched in the future.

Pottery is another type of ceramics, eleven papers indicate that their chosen datasets were potteries, from Japan, Italy, the United States of America, Honduras, France, Egypt, South Africa, and Greece, ranging from the Early Neolithic period (ca. 7850 to 7450 aBP) to the early 19th century (ca. 175 to 125 aBP). Although from different countries, each paper selected only pottery samples from a single period to be studied and classified. One case is under the same period

TABLE 6. Dataset of the reviewed studies.

ID	Туре	Country	Period	Pattern Types	Sample Count	Data Source
ID1 [30]	Porcelain	China	Song dynasty	N/A	N/A	Museum collection
ID2 [31]	Pottery	Japan	Late jomon pe-	N/A	1036 X-ray im-	Not provided
			riod		ages	
ID3 [32]	Porcelain	China	Tang, Song, and	Ewers	232 images	Museum collection
			Yean dynasties			
ID4 [33]	Porcelain	China	N/A	19 types of vases	N/A	Baidu PaddlePaddle
		al i	27/1			
ID5 [26]	Porcelain	China	N/A	N/A	N/A	N/A
ID6 [34]	Porcelain	China	Ming and Qing	Dragon, Fish, Peony, Lotus patterns	N/A	Museum website
1D7 [25]	Doroalain	China	N/A	Various	5924 imagas	Onlina Panasitany
ID7 [35]	Porcelain	China	Vuan Ming and	Shards by incomplete patterns	373 images	Scholarly book
100 [50]	1 Oreerann	Ciina	Oing dynasties	Shards by meonipiete patterns	575 mages	lingdezhen relics
ID9 [37]	Pottery	Italy	Between the Late	Vessel profiles	4992 images	Latium Vetus and Etruria
	,		bronze age and	· · · · · · · · · · · · · · · · · · ·		archaeological sites
			the Orientalizing			6
			period			
ID10 [27]	Porcelain	China	N/A	N/A	N/A	N/A
ID11 [38]	Porcelain	China	Ming and Qing	Mei vase, Yuhuchun vase	248 images	Beijing Museum and
			dynasties			Taipei Museum website
ID12 [39]	Ceramic	Thailand	N/A	7 types of patterns	557 images	Museum collection
ID13 [29]	Porcelain	China	Song, Yuan,	Blue and white porcelain, Purple	2750 images	Palace Museum website
			Ming, and Qing	sand, Doucai		
	Dattamy	LICA	dynasties	Tugayan White Ware shands	2064 imagaa	Mussum collection
ID14 [40]	Pottery	USA	ca.1125 to 050	Tusayan white ware sherds	5064 images	Museum collection
ID15 [41]	Ceramic	Korea		Various	19610 images	Museum collection
ID15 [41]	Porcelain	China	N/A N/A	7 types of bottles	5624 images	Baidu crawled data
ID17 [28]	Porcelain	China	N/A	N/A	N/A	N/A
ID18 [7]	Pottery	Honduras	Cocal period	Various	1069 images	Guadalupe archaeological
					from 66 pottery	site
					sherds	
ID19 [43]	Pottery	France	Early Neolithic	Coiling and Spiral Patchwork	16 sherds	Pendimoun rock shelter
			period			
ID20 [44]	Porcelain	China	Yuan, Ming, and	Banana leaf patterns	230 pieces	Museum collection
			Qing dynasties			
ID21 [45]	Porcelain	China	Ming dynasty	Washer	N/A	Academic research
ID22 [46]	Pottery	USA	Pre-Colonial	Paddle impressions on pottery	N/A	North America archaeo-
1D22 [47]	Dottomi	Equat	Nila alay and	Sherds	60 complac	Indical sites
ID25 [47]	Follery	Egypt	Marl clay	Nine cray and Marrieray	09 samples	Museum Conection
ID24 [48]	Ceramic	New	N/A	Various	N/A	New Zealand sites
	Ceruinie	Zealand		Various		Thew Zeuland Sites
ID25 [49]	Pottery	South	N/A	Zulu pottery fabrics	50 per class	Field collection.
		Africa		1 5		
ID26 [50]	Ceramic	USA	ca.175 to 125	N/A	8 classes of frag-	Independence National
			aBP		ments	Historical Park
ID27 [51]	Porcelain	China	N/A	N/A	288 pieces	Academic research
ID28 [52]	Porcelain	China	Tang, Song dy-	Jars	50 images	Forbidden City Museum
			nasty, Yuan, and			
1000 (50)		TIC 1	Ming dynasties	NT/A		
ID29 [53]	Ceramic	USA	ca.1/5 to 125	N/A	6 pieces of frag-	Independence National
ID20 [14]	Dottomy	Franco	aDP	Wheel correct	057 patterns	Filstofical Park
	rouery	Trance	aBP		257 patterns	Sarah archaeological sile
ID31 [54]	Potterv	Greece	N/A	N/A	203 pottery	Koroneia archaeological
	1 onory				sherds	site
ID32 [55]	Ceramic	USA	ca.175 to 125	N/A	N/A	Independence National
		_	aBP			Historical Park
ID33 [56]	Pottery	Greece	Byzantine era	N/A	110 pottery sherd	Byzantine antiquities
					images	

for pottery with different patterns, such as ID19 [43], which investigated potteries with coiling and spiral patchwork, ID14 [40] that focused on TWW pottery sherds with decorative patterns, Figure 5 shows the samples of dataset in ID14. The other case regards the same period but focuses on different types of pottery, such as ID9 [37] where the authors classified potteries of different vessel profiles. At the same time, most of the datasets are focused on pottery sherds, which is one



FIGURE 4. Distribution of the dynasties in the porcelain datasets.



FIGURE 5. Samples of TWW sherds in ID14 [40].

of the dilemmas faced by archaeological pottery. The manual classification of a large number of sherds is a time-consuming activity [7]. Hence, the main objective of the researchers is to reduce the time of pottery classification and improve the consistency of results.

The remaining six papers, which do not specify the type of ceramics, from Thailand, Korea, New Zealand, and the United States of America and do not specify the period of the dataset, but focus mainly on ceramic patterns, such as ID12 [39], that classified ceramics with seven types of patterns from Sukhothai kiln, Figure 6 shows the samples of dataset in ID12.

2) DATA SOURCE AND SAMPLE COUNT OF CERAMIC

The size of the dataset ranges from tens of samples to thousands of samples, depending on the preservation of the ceramics. For example, in ID2 [31], the number of samples preserved from this period was insufficient and therefore



FIGURE 6. Samples of CMC sukhothai ceramics dataset in ID12 [39].

the labelled data was insufficient, and self-built experimental data had to be used for the study. ID20 [44] has a similar situation with a limited number of available porcelain with banana leaf patterns. Therefore, the dataset size was only 230 samples. Having larger datasets enables the use of powerful algorithms, such as deep learning models, which typically yield superior results. The richness and volume of data allow these algorithms to learn more complex patterns and representations, thereby improving their accuracy and generalization capabilities.

As shown in Table 6 (column Data Source), the majority of the data were collected from the museum collection, which includes both images photographed in the museum and images downloaded from the museum's website, as well as datasets obtained from archaeological excavation sites. Additionally, a small number of datasets were obtained through web crawling (ID4, ID7 and ID16). The diverse methods of data collection ensure a broad and varied dataset. enriching the training material for the models. However, the quality of these datasets, as well as the trustworthiness of the labels of the images, is crucial. For instance, images obtained from official museum sources are likely to have accurate and reliable labels due to the expertise and meticulous documentation practices of museum professionals. Conversely, images from non-official museum websites or those collected via web crawling might have less reliable labels. This variability in label accuracy can significantly impact the training of machine learning models. Accurate labels are essential for the models to learn correctly, as they guide the algorithms in understanding and categorizing the images. Inaccurate labels

can lead to incorrect learning, reducing the effectiveness and accuracy of the trained models. Therefore, ensuring the reliability of the labels, especially in self-built datasets, is vital for developing robust and effective machine learning models.

3) DATA SCARCITY AND FUTURE DIRECTIONS

While ML algorithms have emerged as a promising tool for archaeologists, offering the potential to streamline the identification of archaeological ceramics, they still have to face a significant hurdle, data scarcity. In fact, to be truly effective, machine learning algorithms require a robust dataset of well-documented ceramic artefacts. This data should include detailed information on the origin, type, composition, and visual characteristics of each ceramic piece. Unfortunately, such comprehensive datasets are often scarce in archaeology.

Several factors contribute to data scarcity. On the one hand, archaeological excavations are often resource-intensive, and not all recovered ceramics receive in-depth analysis due to time and budget constraints. However, even if museums possess this data, several challenges remain. In fact, this information often exists in data silos, not readily transformed into usable datasets for training ML algorithms. Cataloguing and documenting ceramics can be a time-consuming and resource-intensive process. Many museums lack the staff or funding to create comprehensive, digital datasets for their entire collections. Finally, even if museums have digitized data, they might not readily share it for free due to concerns about ownership and potential commercial use.

To partially address this challenge, there are several initiatives to establish standardized recording practices for archaeological ceramics, with the aim of building and sharing open-access databases of ceramic data. Finally, the recent advancements in multimodal Large Language Models able to generate images from text inputs could be exploited to generate synthetic data to augment the size of the datasets [57].

C. IMAGE PROCESSING FOR ARCHAEOLOGICAL CERAMICS IDENTIFICATION

This Section aims to answer the second research question **RQ2**, "What are the findings about the image processing of archaeological ceramics for its identification?". Image processing is the process of transforming an image into a digital form and performing certain operations to get some useful information from it [58]. It encompasses a variety of techniques applied to digital images, typically in a sequential order, such as image enhancement, image segmentation, and feature extraction. Table 7 reports on image processing methods of the reviewed studies. Among all the thirty-three papers reviewed, thirty-two (97%) of them employed image processing methods for tasks such as feature extraction [30] and image segmentation [34] of archaeological ceramics. The remaining paper, ID33 [55], did not provide specific information on any image processing method.

1) OVERVIEW OF IMAGE PROCESSING METHODS

Image processing is crucial in the classification of ancient ceramic images and in the reviewed studies, including three main image processing approaches: feature extraction, image segmentation, and image enhancement.

Regarding feature extraction, an image feature is a primitive attribute of an image. Some features are natural and they are defined by the visual appearance of an image, while others are artificial features that result from specific manipulations of an image. Natural features include the luminance of a region of pixels and gray-scale textural regions [59]. Image amplitude histograms and spatial frequency spectra are examples of artificial features. Image features are important in the isolation of regions of common property within an image (image segmentation) and the subsequent identification or labelling of such regions (image classification).

Segmentation of an image involves dividing or separating the image into regions with similar attributes. The most basic attribute for segmentation is image luminance amplitude for a monochrome image and colour components for a colour image. Image edges and texture are also useful attributes for segmentation [60].

Image enhancement is a collection of techniques designed to improve the visual appearance of an image. For example, an image enhancement system might use high-frequency filtering to emphasize the edges of objects in an image [59]. In this application, the image enhancement processor would emphasize salient features of the original image and simplify the processing task of data extraction.

2) FOCUS ON FEATURE EXTRACTION, IMAGE SEGMENTATION, AND IMAGE ENHANCEMENT

Depending on the objectives of the papers, different image processing methods are used. Primarily, they regard feature extraction, image segmentation, image pre-processing, image enhancement, colour space transformation, and image registration. Figure 7 shows the frequency of image processing tasks in the reviewed literature. In some studies, there is not a single task, but a combination of them.

Thirty papers (93.7%) employed a feature extraction method, highlighting the central role of feature extraction in machine learning applications [30]. Different studies took advantage of several feature extraction methods according to different ceramic types (pottery and porcelain). Figure 8 shows the distribution of feature extraction methods in reviewed papers. These methods include traditional approaches, such as GVF and LBP to measure the counter of porcelains (ID3 [32]), GLCM to extract texture features, and HSV to extract color features (ID7 [35]), Tamura texture feature (ID21 [45]). In addition to traditional feature extraction methods, there are shallow ML-based methods. ID27 [51] applied Kernel mean shift clustering to extract features (color, texture, and shape) of porcelain images. Both ID30 [14] and ID31 [54] used BoVW to extract features

TABLE 7. Image processing of the reviewed studies.

ID	Tasks	Image Processing Methods
ID1 [30]	Feature Extraction	Convolutional neural network (CNN) used for feature extraction of porcelain patterns while Otsu's method is
	Image Segmentation	employed for image segmentation.
ID2 [31]	Feature Extraction	This study used the CNN for feature extraction of potteries.
ID3 [32]	Feature Extraction	This study used the gradient vector flow field (GVF) soft contour information and LBP operator to measure spatial
	Image Segmentation	coherence, contour continuity, and texture similarity of porcelain ewer images. Principal Component Analysis (PCA)
	ininge segmentation	was used to reduce the dimensionality. The segmentation was based on the SLIC-Neut
ID4 [33]	Feature Extraction	This study used the CNN for feature extraction of porcelain vases
ID5 [26]	Feature Extraction	This study used gray level co-occurrence matrix (GLCM) to extract texture features of archaeological ceramic
105 [20]	Image Segmentation	surfaces. Employed the CNN for feature extraction of ceramics. Used the watershed segmentation algorithm to
	inage beginentation	segment the certain decoration images
ID6 [34]	Image Segmentation	Four CNN models pyramid scene parsing network (PSPNet) DeenI aby3± dual attention network (DANet) high-
ID0 [34]	image beginemation	resolution network (HRNet) to seement percelain percelain sectors dragen fich lotts and peony
ID7 [35]	Feature Extraction	GLCM and average Euclidean distance of Tamura texture features used as texture features of porcelains. Colour
ID7 [55]	Peature Extraction	bictoriand average Euclidean distance of familia texture reduces used as texture reduces of porcetains. Colour bictorian and hus saturation value (HSV) colour space average similarity rate as colour features
ID8 [36]	Imaga Prancoccesing	instegrant and nuc, saturation, value (mployed as part of the image proprogram and nuc, saturation along the space of the image proprogram and nuc, saturation along the space of the image proprogram and the space space of the space o
ID8 [30]	Image Enhancement	A cool segmentation algorithm is employed as part of the inage preprocessing strategy, converting the cool space from rad graen and blue (PGR) to HSV ECUMix was used as an image anhancement method based on the CuMix
	Feature Extraction	motion red, green, and once (NGB) to risk. Te curvity was used as an image eminance internet assesse on the Curvity matching environment of the curvity matching the set of the curvity in
ID0 [37]	Feature Extraction	Norial and an and the second to compare into a lower dimensional energies and the reconstruct them
ID9[37]	reature Extraction	variational autoencoder (VAE) used to compress images into a lower-omnensional space and then reconstruct them, thereby learning to avised the assembled compress images into a lower-omnensional space and then reconstruct them,
ID10[27]	Easture Extraction	ulticity feating to extract the essential features from the poterty profiles.
ID10[27]	reature Extraction	disortioning derivative approximation operator used to detect the edges of the copied image, Zernike moment
ID11[20]	Imagaa Duanna aaaaina	uscretization for realitie extraction.
ID11[56]	Easture Extraction	Data augmentation was achieved by a series of mage preprocessing methods, totation, translation, adjustment of
ID12 [20]	Feature Extraction	chromatic aberration, amplification, distortion, and increase of noise. CNN used for feature extraction.
ID12 [39]	Image Ennancement	image enhancement included rotation, width shift, and zoom. Employed five CNN models (DenseNet121, incep-
ID12 [20]	Feature Extraction	tion v5, vGr16, GoogLeNet, and AlexNet) for feature extraction of patterns from ceramics.
ID13 [29]	Not provided	Not provided
ID14 [40]	Feature Extraction	Employed two CNN models(VGG16, ResNet50) for feature extraction of patterns from potteries
ID15 [41]	Image Segmentation	Mask R-CNN used for segmentation to classify the ceramic form and generate the mask of the ceramic colours.
ID1(142)	Feature Extraction	Applied CNN model inception V2 to extract features of ceramics according to their forms, materials, and patterns.
ID16 [42]	Feature Extraction	This study used VGG16 for feature extraction of porcelain bottles
ID17 [28]	Image Preprocessing	Image quality improved using geometric distortion correction, image enhancement, and smoothing filtering. Edge
	Image Segmentation	detection is employed to define the shape and contours of the ceramics; and threshold segmentation for differences
ID10 [7]		in grey characteristics between the ceramics and the background to produce a binary image.
ID18[/]	Image Enhancement	Image enhancement including random rotation, horizontal and vertical hipping, and random image brightness
10101421	Feature Extraction	adjustments. Employed two CNN models (VGG19, ResNet50) for feature extraction of pottery fabrics.
ID19 [43]	Image Segmentation	Image Segmentation for segment pore regions. Applied Hough Transform for feature extraction to investigate pores
1020 [44]	Feature Extraction	of potteries. Employed CNN model (GoogleNet) for feature extraction of segmented images
ID20 [44]	Easture Extraction	image preprocessing including grayscale processing, KOI region detection, size normalization, inter denoising for
ID21 [45]	Feature Extraction	improve quarty of patients. Applied LDF and HOO fusion feature extraction of banana real patients.
ID21 [45]	Imaga Sagmantation	Eight-chain code menod combined with a watershed angoint in such to the shape extraction. The texture feature
ID22 [46]	Image Segmentation	was extracted by fainting methods, inscriptions segmented into single nandwritten clinices characters.
ID22 [40]	Image Segmentation	Extract curve patterns from a sherd using an FCN-based image segmentation method. A dual-source CNN was
	Feature Extraction	applied for feature extraction. Chamfer matching algorithm to match the pottery sherds to find their underlying
1022 [47]	Eastern Esteration	uesigns
ID25 [47]	reature Extraction	combination of three texture reactives extraction methods: I) Gabot miles for local and offened ones, ii) Laws
1024 [49]	Interne Descieturation	texture for levels, edges, spots, waves, and ripples; m) Haraffeck stexture features for the distribution of grey levels.
ID24 [48]	Easture Extraction	image registration includes template matching algorithms for search for a specific patient within a larger set of data
ID25 [40]	Feature Extraction	according to the variations in scale, rotation, and corour. Enproyee deep neural network for realistic extraction.
ID25 [49]	reature Extraction	Applied weighted neighbour distance using compound metachy of agoittmins representing morphology to extract features from pottery images including statistical textural and pottery features.
ID26 [50]	Color Space Trans	Caranis from pottery mages including statistical, textural, and pattern reduces.
1D20 [50]	formation	cub lacks and the features averaged from Kers on blacks were here divided into six equal-sized
	Feature Extraction	sub-blocks and the reatiles extracted from these sub-blocks were based on the ris v values.
ID27 [51]	Feature Extraction	Applied kernel mean shift clustering to extract features (colour texture, and shape) of porcelain images
ID27 [51]	Feature Extraction	Applied which mean similer distorting to extract relatives (colour, texture, and single) of portection images. Employed LBD to avtract features: (a multi-schemal colour, texture, and single) of portection images.
1020 [32]	reature Extraction	colour space is utilized to characterize the glaze colours of the porcelaine
1D20 [53]	Color Space Trans-	The images of caramic fragments were converted from RGB to HSV colour space. GLCM was applied to extract the
1029 [55]	formation	The mages of ceramic magnetic converted from KOB to TISV colour space. Obevirwas applied to extract the tayling feature of ceramic images
	Feature Extraction	texture reature of ceranic images.
ID30 [14]	Feature Extraction	Cabor filters were applied for extraction of descriptors while Bag of Visual Words (BaVW) extracted visual words
1030 [14]	i cature Extraction	based on the local image descriptors
ID31 [54]	Feature Extraction	Body on the local mage descriptors. Body extracted features on the local image descriptors: colour and texture features were extracted from potters:
LJ1 [J4]	r cature Extraction	image
ID32 [55]	Feature Extraction	Puramid histogram of visual words (PHOW), which is an avtancian of DaVW mode by avtracting local features
m32 [33]	Feature Extraction	from different levels of a puremid structure and then quentizing these features into a visual workshort.
ID33 [56]	Fastura Extraction	To mean under the restrict of a pyramic structure and then quantizing these realities mite a visual vocabiliary.
[06] 6641	I caluit Extraction	but was used to detect edges in the shere's
	mage beginemation	account was used to detect edges in the shelds.



FIGURE 7. Frequency of image processing tasks in reviewed literature.



FIGURE 8. Distribution of feature extraction methods in reviewed papers.

from pottery images. Finally, thirteen papers employed deep neural networks for feature extraction including VAE model [37] and CNN model such as ResNet [36], AlexNet [39], VGG16 [42]), etc. From Figure 8, it can be seen that the CNN model for feature extraction is the most used method since DL has proven to be effective in feature extraction [30] and CNN represents the state-of-the-art in many image-related tasks [61].

Ten papers employed image segmentation techniques. Figure 9 shows the distribution of such methods. ID1 [30] used Otsu's method for image segmentation of porcelains, ID3 [32] applied SLIC-Ncut algorithm to segment porcelain ewers. Finally, ID6 [34] used four CNN models to segment porcelain patterns while ID15 [41] employed Mask R-CNN for ceramics segmentation.

Finally, three papers employed image enhancement techniques. ID8 [36] used FCutMix which is an image enhancement method based on DL, ID12 [39] and ID18 [7] used traditional enhancement methods including random rotation, horizontal and vertical flipping. Two papers indicated image registration such as Chamfer matching algorithm [46] and



FIGURE 9. Distribution of image segmentation methods in reviewed papers.

template matching algorithm [48] for matching the pottery sherds or patterns. The other two papers (ID26 [50] and ID29 [53]) discussed color space transformation to convert RGB color format to HSV color format.

3) EVOLVING LANDSCAPE OF IMAGE PROCESSING METHODS

The previous Subsection highlighted that CNNs are the most used algorithms for both feature extraction and image segmentation. Figure 10 shows the evolving trend in the use of CNNs for these two tasks. Especially from 2021, CNNs became the dominant method for image processing in the field of archaeological ceramic images.

Such a shift in terms of popularity from traditional image processing methods (LBP, GLCM, Tamura texture feature, etc.) and traditional ML-based methods (Kernal mean shift clustering) to DL ones, reflects the same behaviour that can be observed in other research areas based on image analysis. In fact, traditional approaches rely on well-defined features of ceramics, and manual feature design is considered time-consuming and inflexible. On the other hand, DL overcome this limitation of handcrafted feature processing [30].

Although CNN's feature extraction method improves the accuracy of extraction, feature extraction of ancient ceramics still faces some challenges, including the inconsistency of surface quality and the complexity and diversity of texture [26]. A promising research direction that could be exploited regards the use of Transformers for feature extraction.

In fact, parallel to what is happening in computer vision [62], Transformers could also be exploited in image analysis of archaeological ceramics. These models excel at capturing long-range dependencies within data [63], a crucial aspect for analyzing complex visual details in ceramics. Unlike traditional methods that focus on local features, transformers can identify subtle relationships between different parts of the image, potentially revealing characteristics like



FIGURE 10. Trend of CNN model methods in feature extraction and image segmentation.

decorative patterns, specific glazing techniques, or even signs of wear and tear. This deeper understanding of the image could lead to a more accurate identification of ceramic origins, types, and potential historical context. Furthermore, transformers can be trained on diverse datasets encompassing a wide range of ceramics, potentially overcoming limitations caused by data scarcity in this field.

D. ML ALGORITHMS FOR THE IDENTIFICATION OF ARCHAEOLOGICAL CERAMICS

This Section answers the third and final research question **RQ3**, "What are the main ML algorithms used in the identification of archaeological ceramics?". The algorithms employed for the identification of archaeological ceramics mainly consist of classification and object detection algorithms [64]. Table 8 reports on the various ML algorithms employed. Among thirty-three papers, twenty-eight (84.8 %) of them use ML algorithms. Among these twenty-eight papers employing ML algorithms, seventeen of them (60.7%) used DL algorithms to identify archaeological ceramics, using various CNN architectures: VGG16, ResNet50, DenseNet121, InceptionV3, GoogLeNet, and AlexNet [38], [40], [42], [43]. The papers employing traditional ML algorithms to classify archaeological ceramics took advantage of SVM, KNN, and SOM [49], [50], [53].

The remaining ones (15.1%), instead, are primarily focused on datasets and image processing methods of the archaeological ceramics and did not conduct experiments for the classification or identification of archaeological ceramics.

1) APPLICATION OF ML ALGORITHMS IN ARCHAEOLOGICAL CERAMIC IMAGES

Figure 11 shows the distribution of applications based on ML/DL algorithms in reviewed papers. Among twenty-eight papers, twenty-five of them applied ML/DL algorithms to image classification, three of them focused solely on image segmentation, and two of them discussed image processing and object detection respectively. It is worth noting that



FIGURE 11. Distribution of applications in reviewed papers.

some papers have more than one application, but combine image segmentation with another application. Anyway, these distributions highlight that the problem of archaeological ceramic identification is often modelled and tackled as a classification one.

Fourteen studies employed a DL algorithm, such as VAE [37] and CNN like VGG [31], ResNet [33], Efficient-Net [31], MobileNet [33], GoogleNet [39], etc., while thirteen studies took advantage of an ML algorithm, including [53], SVM [49], SOM [50], KNN [53], etc.

Although there is a similar amount of reviewed literature based on ML and DL, they were used in different periods. Figure 12 shows the tendency of ML/DL algorithms for image classification. There's a notable increase in the use of DL algorithms starting around 2022, as indicated by the sharp upward line, while ML algorithms trend line shows fluctuations over the years, with peaks and troughs, indicating variable usage or effectiveness during that period. It is interesting to notice that after 2021, ML algorithms are mainly used as baseline models for comparison. Hence, we can conclude that DL algorithms for the classification of archaeological ceramic images have been the dominant trend in the last three years.

Finally, the only paper about object detection is also based on the DL algorithm. It combines object detection with image segmentation, proposing a new tool to allow users to analyze the visual elements of ceramics using their smartphones and devised a method to extract representative colors of ceramics for object detection.

2) PERFORMANCE EVALUATION

The performance evaluation of algorithms for image classification, image segmentation, and object detection is mainly based on evaluation metrics including accuracy, precision, recall, and F1 score [41].

With regard to image classification, the papers based on CNNs architectures perform differently on various datasets. In the reviewed literature, the performance evaluation of the

TABLE 8. ML algorithms of the reviewed studies.

ID	Application	ML Algorithms
ID1 [30]	Image Classification	This study employed a convolutional neural network (CNN) as a primary tool for porcelain image classification with an accuracy of 94.2%.
ID2 [31]	Image Classification	The study employed CNN for image classification of potteries.
		Evaluated the performance of three different CNN architectures: Visual geometry group (VGG), Residual neural
		network (ResNet), and EfficientNet, achieving accuracy of 66%, 77%, and 79%, respectively.
ID3 [32]	Image Classification	This study employed logistic regression cross-validation (CV) for the image classification with an accuracy of 90.7%.
ID4 [33]	Image Classification	The study employed CNN for image classification of porcelain vases.
	-	Compared and evaluated the accuracy and loss values under different optimization parameters under three different
		CNN models: ResNet, Inception, and MobileNet, achieving an accuracy of 93.8%, 91.2%, and 94.2%, respectively.
ID5 [26]	Image Classification	The study employed CNN to extract texture features of ceramics for image classification with an accuracy of 76.3%.
ID6 [34]	Image Segmentation	This paper was based on four CNN models PSPNet, DeepLabv3+, DANet, HRNet to segment porcelain patterns:
		dragon pattern, fish pattern, lotus pattern and peony pattern. Results show that the image segmentation methods of
		HRNet and DANet models are better than the other two models.
ID7 [35]	Not provided	Not provided
ID8 [36]	Image Classification	This study employed a Feature Fusion Convolutional Network (FFCNet), this architecture extends the typical
		capabilities of the standard CNN (ResNet) by incorporating mechanisms specifically designed for effective feature
		fusion, improving the recognition and classification accuracy from 71.7 to 79.5%.
ID9 [37]	Image Processing	Convolutional VAE was designed to work in an unsupervised learning style, suitable for this study where labeled data
		was not available. VAE extracted the essential features of archaeological pottery profiles. The model demonstrated
ID10[27]	Netwarded	its effectiveness and was validated through visual inspection of the reconstructed images against the original profiles.
$\frac{\text{ID10}[27]}{\text{ID11}[28]}$	Not provided	Not provided
ID11 [38]	Image Classification	This paper employee CNN for image classification of porcelain vases, accuracy of the network in identifying the
ID12 [20]	Image Classification	types and ages of allefelt certaints was above 20.4%. This study ampland and compared the performance of five CNN models/DenseNot121_IncentionV3_VCC16_
ID12 [39]	inage Classification	This study employed and compared une performance of the CNN models(Denserver121, inception v3, vGG16, activity) the comparison of the CNN performance showed that the VGG16 activity de bot
		results with accuracy of 5 %
ID13 [29]	Image Classification	This study compared MI algorithm logistic regression for DL algorithm based on transfer learning and found that
1015 [27]	inage classification	the accuracy of DL algorithm was 70% and the accuracy of logistic regression was 48% the DL algorithm improved
		classification results by more than 20% over the logistic regression.
ID14 [40]	Image Classification	This research presented the evaluation for image classification of decorated potteries using two CNN models(VGG16
		and ResNet50) with an accuracy of 82.5%.
ID15 [41]	Image Segmentation	This article employed Mask R-CNN for segmentation to classify the ceramic form and generate the mask of the
	Object Detection	ceramic, the average accuracy of Mask R-CNN was 79.5%.
		For the analysis of material and pattern, the Faster R-CNN was applied by using the region proposal networks (RPN),
		the average accuracy of Faster R-CNN was 60%.
ID16 [42]	Image Classification	This paper compared two different models for image classification: a DL model (VGG16) with an accuracy of 83.9%,
		an integration of DL and ML model (VGG16-GSSVM) with an accuracy of 89.3%
ID17 [28]	Not provided	Not provided
ID18 [7]	Image Classification	This research presented the evaluation for image classification of pottery fabrics in thin sections using two CNN
10101421		models (VGG19 and ResNet50) and the accuracy remained above 95%.
ID19 [43]	Image Classification	This study compared ML method SVM with DL method CNN model (GoogleNet) for image classification. The CNN method thickness them 000°) showed mean security that the SVM exclass them 900°).
ID20 [44]		The CNN method (migher than 90%) showed more accurate than the SVM one (less than 80%).
ID20 [44]	Image Classification	I his study employed SVM combined with LBP-HOG fusion feature extraction for the porcelain image classification of these dynastics achieved 75% of Viso dynastic 90% of Viso dynastic 90% of Origo 100% of Origo 100
ID21 [45]	Not provided	In the equilables, achieved 75% of Tuan dynasty, 60% of Wing dynasty and 96% of Ging dynasty.
$\frac{1D21}{1D22}$ [45]	Image Segmentation	This paper used an ECN based image segmentation method to extract curve patterns from a shard
ID22 [40]	Image Classification	A dual-source CNN (AlexNet) was applied for feature extraction and the segmented curve patterns could be
	ininge chassification	implicitly classified through this method.
ID23 [47]	Image Classification	This paper employed k-Nearest Neighbours (KNN) algorithm for image classification, compared three different
	ininge ensementen	methods for texture extraction. Gabor filters with an accuracy of 68.1% Laws' texture measures with an accuracy of
		73.9%, Haralick's texture features with an accuracy of 71.1%.
ID24 [48]	Image Classification	This study employed a deep neural network for image classification of ceramics.
ID25 [49]	Image Classification	The classification in Wndchrm involves different ML algorithms(decision trees, SVM, or simple statistical compar-
	-	isons), integrated within its framework to classify pottery images.
		The classification accuracy reached between 90% to 95%.
ID26 [50]	Image Classification	This paper applied a Self-Organizing Map(SOM), a type of unsupervised ML model to group the fragments with an
		accuracy of 89.6%.
ID27 [51]	Image Classification	This paper employed semi-supervised mean shift clustering, an ML method for image classification with an accuracy
		of 75.6%
ID28 [52]	Not provided	Not provided
ID29 [53]	Image Classification	This paper employed the KNN algorithm for image classification with an accuracy of 86.5%.
ID30 [14]	Image Classification	SVM was used to classify different patterns of pottery shards using features extracted from Gabor filters and features
		generated by Bags of visual words.
ID21 [54]	Image Classification	Kesuits of classification were presented by a confusion matrix.
ID31 [54]	inage Classification	Compared live different ML algorithms for classification, SVM, KNN, Naive Bayes, Sequential Minimal Optimiza-
ID22 [55]	Imaga Classification	uon (SNO) and Shiple Logic, SVM achieved une accuracy (0.93/30%).
ID32 [33]	Image Classification	Employed two will algorithms, KINN and 5 VIVI on image classification of patterns from ceramics.
[00] ככתו	mage Classification	1 is the and 5 vite used for the classification of patients from pottery sheres.



FIGURE 12. Tendency of ML/DL algorithms on image classification.

ABLE 9. Accurac	y of DL algorithms	on image classification.
-----------------	--------------------	--------------------------

ID	Ceramic Type	Number of Classes	DL Algorithms	Accuracy
ID1 [30]	Porcelain (Yaozho kiln)	N/A	CNN model	94.2%
ID2 [31]	Pottery	7 classes	VGG ResNet EfficientNet	66% 77% 79%
ID4 [33]	Porcelain	19 types of vases	MobileNet ResNet Inception	94.2% 93.8% 91.2%
ID5 [26]	Porcelain	N/A	CNN model	76.3%
ID8 [36]	Porcelain	14 types	ResNet FFCNet	71.7% 79.5%
ID11 [38]	Porcelain	2 types of vases	CNN model	96.4%
ID12 [39]	Ceramic (Sukhothai kiln)	7 types of patterns	VGG16	86.5%
ID13 [29]	Porcelain	3 glaze colours	CNN model	70%
ID14 [40]	Pottery	8 design classes	VGG16, ResNet50	82.5%
ID16 [42]	Porcelain	7 types of bottles	VGG16 VGG16- GSSVM	83.9% 89.3%
ID18 [7]	Pottery	5 types	VGG19, ResNet50	93%
ID19 [43]	Pottery	2 types of patch- work	CNN model	90%
ID22 [46]	Pottery	98 non- composite patterns 20 com- posite patterns	AlexNet	N/A
ID24 [48]	Ceramic	7 classes	CNN model	N/A

models is mainly based on accuracy. Table 9 reports the accuracy of DL algorithms in the classification of ancient ceramics. It can be seen that the classification accuracy is more than 70% in both pottery and porcelain, and the accuracy of ID1, ID4, ID11, ID18, and ID19 even exceeds 90%.

 TABLE 10. Accuracy of ML algorithms on image classification.

ID	Ceramic	Number	ML	Accuracy
	Туре	of	Algorithms	
		Classes		
ID3 [32]	Porcelain	3 dynas-	Logistic	90.7%
		ties	regression	
			cross-	
			validation	
ID13 [29]	Porcelain	3 glaze	Logistic	48%
		colours	regression	
ID19 [43]	Pottery	2 types	SVM	80%
		of patch-		
		work		
ID20 [44]	Porcelain	3 dynas-	SVM	75%
		ties		
ID23 [47]	Pottery	2 classes	KNN	68.1%
ID25 [49]	Pottery	2 types	SVM	90%
	5	of clays		
ID26 [50]	Ceramic	8 classes	SOM	89.6%
		of frag-		
		ments		
ID27 [51]	Porcelain	13	Mean shift	75.6%
		classes	clustering	
ID29 [53]	Porcelain	6 classes	KNN	86.5%
		of frag-		
		ments		
ID30[14]	Pottery	6 classes	SVM	N/A
ID31 [54]	Pottery	N/A	SVM	93.96%
ID32 [55]	Ceramic	6 classes	KNN, SVM	N/A
		of frag-	· ·	
		ments		
ID33 [56]	Pottery	2 classes	KNN, SVM	N/A
		of		
		sherds		

The achieved results are in line with the state-of-the-art image classification using deep learning algorithms.

Table 10 reports the accuracy of ML algorithms in image classification. Seven papers employed SVM and achieved accuracy higher than 70%. In ID13 and ID19, ML and DL algorithms were applied separately to image classification to compare the results. They showed that the accuracy of DL algorithms was higher than the on achieved by ML algorithms. Although SVM has achieved good classification results as an ML algorithm, for example on pottery shards, its performance was still inferior to that of the DL algorithm on the same dataset [43]. Therefore, the choice of the DL algorithm as an image classification of archaeological ceramics appears to be a natural choice.

E. LIMITATIONS OF THIS REVIEW

To address the traditional limitations of systematic reviews, we followed the PRISMA methodology in this work. However, this approach still presents several limitations. Firstly, the review process was highly selective, focusing only on scientific articles published in journals or conference proceedings from reputable databases to ensure high-quality content. Secondly, the study of citations or reference analysis was deliberately excluded in order to keep the paper more concise and focused. However, future research efforts aimed at a more comprehensive overview could expand the range

TABLE 11. Details of the reviewed studies.

ID	Paper Title	Published Year
ID1 [30]	Technological innovation in the recognition process of Yaozhou Kiln ware patterns based on image classification	2023
ID2 [31]	Classification of unexposed potsherd cavities by using deep learning	2023
ID3 [32]	Identification of porcelain ewers in Tang, Song, and Yuan dynasties by digital shape characterization	2023
ID4 [33]	Comparative Study on Image Classification Models of Ancient Ceramic Types	2023
ID5 [26]	Ceramic Texture Extraction Method Based on Convolution Neural Network and Image Processing	2023
ID6 [34]	Typical Ceramic Pattern Extraction Based on Convolutional Neural Network	2023
ID7 [35]	Using Image Feature Extraction to Identification of Ancient Ceramics Based on Partial Differential Equation	2022
ID8 [36]	Automatic Classification of Blue and White Porcelain Sherds Based on Data Augmentation and Feature Fusion	2022
ID9 [37]	A deep variational convolutional Autoencoder for unsupervised features extraction of ceramic profiles. A case study from central Italy	2022
ID10[27]	Ceramic Decoration Extraction Method Based on Computer Vision and Image Processing	2022
ID11[38]	Intelligent Dating of Chinese Ancient Ceramics Based on Convolutional Neural Network	2022
ID12 [39]	Using Deep Learning for the Image Recognition of Motifs on the Center of Sukhothai Ceramics	2022
ID13 [29]	Ceramic Type Recognition Algorithm Based on Ontology Modeling and Transfer Learning	2022
ID14 [40]	Applications of deep learning to decorated ceramic typology and classification: A case study using Tusavan White	2021
	Ware from Northeast Arizona	
ID15 [41]	Smart Culture Lens: An Application That Analyzes the Visual Elements of Ceramics	2021
ID16[42]	Identification Method of Ancient Ceramics Revision	2021
ID17 [28]	The Application of Computer Vision in the Detection of Ancient Ceramic Art	2021
ID18 [7]	Ceramic Fabric Classification of Petrographic Thin Sections with Deep Learning	2021
ID19 [43]	Applications of learning methods to imaging issues in archaeology, regarding ancient ceramics manufacturing	2021
ID20 [44]	Characteristics of Banana Leaf Pattern for Ceramics in Yuan, Ming and Qing Dynasties	2020
ID21 [45]	Research on ancient ceramics identification by artificial intelligence	2019
ID22 [46]	A Framework for Design Identification on Heritage Object	2019
ID23 [47]	Automated classification of archaeological ceramic materials by means of texture measures	2018
ID24 [48]	Machine learning identification and classification of historic ceramics	2018
ID25 [49]	Using supervised machine learning to classify ceramic fabrics	2018
ID26 [50]	Classification Archaeological Fragments into Groups	2017
ID27 [51]	Porcelain image classification based on semi-supervised mean shift clustering	2017
ID28 [52]	Machine Vision Based Classification and Identification for Non-destructive Authentication of Ancient Ceramic	2017
ID29 [53]	Using Both HSV Color and Texture Features to Classify Archaeological Fragments	2015
ID30[14]	Automatic pattern recognition on archaeological ceramic by 2D and 3D image analysis: A feasibility study	2015
ID31 [54]	Towards the automatic classification of pottery sherds: two complementary approaches	2015
ID32 [55]	A Semantic Analysis Approach to Thin-Shell Ceramic Fragment Classification	2013
ID33 [56]	Automatic classification of archaeological pottery sherds	2013

of review sources and employ knowledge graphs to visually describe developing tendencies and research hot spots in the application of machine learning algorithms to archaeological ceramics identification in a more sophisticated way. Thirdly, this literature also had unaddressed research areas, first was integration with other technologies, the potential integration of DL algorithms with other technologies such as 3D scanning and VR for more comprehensive analysis of archaeological ceramics, second was cross-cultural comparative studies, investigating how DL algorithms can be used to the identification of archaeological ceramics across different cultures and historical periods more effectively. Finally, since this research began at the beginning of 2024, it includes reviewed studies that have been published during the last ten years (i.e., between 2013 and 2023). The limitation of this timeframe indicates that future works could combine studies from 2024 and subsequent years, which will have the potential to provide deeper insights and more varied applications for the review of archaeological ceramics.

V. CONCLUSION

The purpose of this paper is to examine the trends of machine learning in the identification of archaeological ceramics. Specifically, it focuses on the application of ML and DL algorithms between 2013 and 2023, on several types of ceramic (pottery and porcelain). It investigates several aspects such as the available datasets, the image processing methods, and ML or DL algorithms.

In terms of the application of ML algorithms to the image classification of ceramic images, papers from 2013 to 2020 tend to use traditional ML algorithms such as SVM and KNN. However, there is a general shift towards the use of DL based on CNN models for classification in papers from 2020 to 2023. In fact, based on the conclusions drawn from these papers, DL algorithms have made significant breakthroughs in many tasks due to their powerful automatic feature extraction capabilities. Among CNN models, VGG16, MobileNet and ResNet have achieved good performance on archaeological ceramic images and higher accuracy compared to traditional ML algorithms such as SVM and KNN. This indicates that DL algorithms such as CNN models will be the main trend in the future for archaeological ceramic classifications.

Various studies demonstrated the technical potential of DL for the identification of archaeological ceramics. The manual classification of a large number of pottery sherds is a time-consuming activity and empirical identification methods of porcelains often lack objective constraints. The

TABLE 12. Dataset details of the reviewed studies.

ID	Ceramic Type	Dataset
ID1 [30]	Porcelain	This study analyzed Yaozhou kiln ceramic from China. The production period is not provided.
ID2 [31]	Pottery	This study analyzed archaeological pottery fragments from the Late Jomon period (ca.4000 to 3200 aBP) in Japan.
		The proposed dataset is composed of 1,036 X-ray images from seven classes.
ID3 [32]	Porcelain	This study selected images of porcelain ewers from the Tang (ca.1332 to 1043 aBP), Song (ca.990 to 671 aBP),
		and Yuan (ca.679 to 582 aBP) dynasties in China. There were 85 images from the Tang Dynasty, 85 from the Song
		Dynasty, and 62 from the Yuan Dynasty. Images were photographed at museums in China.
ID4 [33]	Porcelain	The study used the porcelain image dataset provided by Baidu PaddlePaddle to collect archaeological porcelain
ID 5 (2(1		images in the datasets, including 19 different types of vases in China. The production period is not provided.
ID5 [26]	Ceramic	Not provided
ID6 [34]	Porcelain	The datasets are mainly derived from Chinese museums photo scans and websites. Four typical porcelain pattern detector of dragen fight lates and nearest pattern activities devices and websites.
ID7 [25]	Doroalain	datasets of dragon, fish, fotus, and peonly are established. The production period is not provided.
ID7 [55]	Forcelain	The study extracted and valuated 5,534 mages of alchaeological centaries from China, including 272 types of Kins, Vacybou kiln, and Vae kiln, with manual labelling. The production period is not provided
ID8 [36]	Porcelain	Table no the second sec
100 [50]	Toreclam	of 14 types in the Yuan (ca 679 to 582 aBP) Ming (ca 582 to 306 aBP) and Oing (ca 306 to 38 aBP) dynasties from
		China.
ID9 [37]	Potterv	This study focused on Lazio and Etruria pottery, a dataset of 4992 complete vessel profiles from 117 sampled
		contexts between the Late Bronze Age and the Orientalizing period (ca.3100 to 2530 aBP) in Italy.
ID10[27]	Ceramic	Not provided
ID11 [38]	Porcelain	This paper takes Mei vase and Yuhuchun vase in Ming (ca.582 to 306 aBP) and Qing (ca.306 to 38 aBP) dynasties as
		the research object, all samples are from the Beijing Museum and Taipei Museum website. A total of 248 available
		samples were obtained.
ID12 [39]	Ceramic	The images in the dataset were obtained by photographing the ceramics of Sukhothai kilns in many private museums.
		A total of 557 images that featured seven types of patterns were collected from Thailand.
ID12 [20]	Describe la	Not provided the production period of the ceramics.
ID13[29]	Porcelain	In dataset of the paper was collected from the oriclal website of the Palace Museum in China. A total of 2/300 pictures of the most paragraphic according color along only and the palace blue only white paragraphic according to the palace blue only white palace blue only white paragraphic according to the palace blue only white palace blue only white paragraphic according to the paragraphic according to the palace blue only white palace blue only white paragraphic according to the p
		and durcai from Song (cg. 900 to Cri anto, vino Cri and cri an
		to 38 aBP) dynasties
ID14 [40]	Pottery	A set of 3064 Tusavan White Ware(TWW) sherd photographs for the dataset from regional museum collections in
	1 0 0000 1 9	the United States of America (ca.1125 to 650 aBP), providing a rich source of imagery due to the contrast between
		the paint and background of the sherds.
ID15 [41]	Ceramic	The dataset required for this study was created using two approaches. First, A total of 5,684 images of 5,211 ceramics
		were collected by the National Museum. Second, 13,926 Pictures of 2,135 ceramics were selected at 19 institutions
		in the Republic of Korea. Not provided the production period of the ceramics.
ID16[42]	Porcelain	The self-built porcelain set includes seven types, 5624 pieces of porcelain bottles which had been crawled through
ID17 [20]	C	Baidu in China. Not provided the production period of the porcelains.
ID17[28]	Pottery	Not provinced
ID18[/]	rottery	cross-polarized light from the Cocal-period archaeological site of Guadalune from Honduras (ca 950 to 425 aBP)
		In total 1069 images were used to train each model.
ID19 [43]	Pottery	The dataset selected in this study was composed of sixteen sherds: four coiling technology (CoT) sherds and twelve
		spiralled patchwork technology (SPT) sherds. The dataset originates from the early Neolithic period (ca.7850 to
		7450 aBP) in southeastern France.
ID20 [44]	Porcelain	A total of 230 pieces of ancient porcelains containing banana leaf patterns from Yuan (ca.679 to 582 aBP), Ming
		(ca.582 to 306 aBP) and Qing (ca.306 to 38 aBP) dynasties, collected from museums in China.
ID21 [45]	Porcelain	Analyzed the porcelain washer from Ming dynasty (ca.582 to 306 aBP) in China.
ID22 [46]	Pottery	The dataset consists of curvilinear paddle impressions on pottery sherds and comes from North America, specifically during the agrid (on 1600 to 1200 to 200).
ID22 [47]	Dottom	during the period (ca.1000 to 1500 aBP)
ID25[47]	rottery	types of Equiption clays: Nile clay(ca 2500 to 2450 aRP) and Marl clay (ca 2500 to 3450 aRP). A total of the 60
		samples were imaged under the ordical microscope as this sections using const-old rized light.
ID24 [48]	Ceramic	The dataset includes historic ceramics recovered from archaeological sites in New Zealand. Not provided the
		production period of the ceramics.
ID25 [49]	Pottery	The dataset used in the study consists of pottery samples from Zulu communities in South Africa. The potteries are
		contemporary, made by modern potters who employ traditional methods that are reflective of historical practices.
ID26 [50]	Ceramic	This study selected eight classes of the Independence National Historical Park dataset(ca.175 to 125 aBP), a
1027 [51]	D I :	collection of ceramic fragments in the United States of America.
ID27[51]	Porcelain	Dataset images consist of 288 porcelain pieces, including 13 different porcelains from China. Not provided the
1D28 [52]	Porcelain	A total of 50 percelain increases were selected from Forbidden City Museum of China, including five ages of
ID20[52]	Toreclain	Neolithic (ca 12000 to 4000 aBP) Tang (ca 1332 to 1043 aBP) Song (ca 990 to 671 aBP) Yuan (ca 679 to 582 aBP)
		and Ming (ca.582 to 306 aBP).
ID29 [53]	Ceramic	This study selected six images of the Independence National Historical Park dataset (ca. 175 to 125 aBP), a collection
		of ceramic fragments in the United States of America.
ID30[14]	Pottery	The dataset used includes 935 shard imprints manually obtained by archaeologists (ca.1524 to 924 aBP) from France.
ID31 [54]	Pottery	A set of 203 pottery sherds collected from Greece. Not provided the production period of the potteries.
ID32 [55]	Ceramic	This paper selected ceramic images of the Independence National Historical Park dataset (ca.175 to 125 aBP), a
	_	collection of thin shell ceramic fragments in the United States of America.
ID33 [56]	Pottery	A set of 110 pottery sherd images including six different classes. The sherds were selected and photographed from
		Byzantine Antiquities in Greece (ca.1330 to 330 aBP).

image classification based on DL provides support for the classification of archaeological ceramics. This review and its conclusions should be seen as a first step in the analysis of machine learning for ceramic identification in the sector.

Overall, the use of machine learning algorithms in archaeological ceramics classification complements traditional methods by offering efficiency, accuracy, and new avenues for data analysis and interpretation. By combining the strengths of both approaches, researchers can gain deeper insights into past civilizations and cultural practices. The efficiency of these algorithms can speed up the classification process allowing archaeologists to analyze larger datasets and automate repetitive tasks, such as sorting and categorizing artifacts based on specific attributes. This automation frees up archaeologists' time to focus on higher-level analysis and interpretation. Moreover, machine learning algorithms can uncover hidden relationships or correlations within ceramic datasets that may not be apparent through traditional methods. By analyzing these insights, archaeologists can gain a deeper understanding of cultural practices, trade networks, and historical contexts.

ABBREVIATIONS

AI:	Artificial Intelligence.
BoVW:	Bag of Visual Words.
CNN:	Convolutional Neural Network.
DANet:	Dual Attention Network.
DL:	Deep Learning.
FCN:	Fully Convolutional Network.
FFCNet:	Feature Fusion Convolutional Network.
GLCM:	Gray Level Co-occurrence Matrix.
GVF:	Gradient Vector Flow Field.
HOG:	Histogram of Oriented Gradient.
HRNet:	High-Resolution Network.
HSI:	Hue, Saturation, Intensity.
HSV:	Hue, Saturation, Value.
KNN:	K Nearest Neighbor.
LBP:	Local Binary Patterns.
ML:	Machine Learning.
PCA:	Principal Component Analysis.
PSPNet:	Pyramid Scene Parsing Network.
PRISMA:	Preferred Reporting Items for Systematic
	Reviews and Meta-Analyse.
ResNet:	Residual Neural Network.
RGB:	Red, Green, and Blue.
RPN:	Region Proposal Networks.
SOM:	Self-Organizing Map.
SSD:	Single Shot Detection.
SVM:	Support Vector Machine.
TWW:	Tusayan White Ware.
VAE:	Variational Autoencoder.
VGG:	Visual Geometry Group.
VR:	Virtual Reality.
Wndchrm:	Compound hierarchy of algorithms repre-
	senting morphology.
YOLO:	You Only Look Once.

APPENDIX

See Tables 11 and 12.

ACKNOWLEDGMENT

The support provided by the China Scholarship Council (CSC) during the Ph.D. research of Ziyao Ling at the University of Bologna is acknowledged.

REFERENCES

- L. Ruck and C. T. Brown, "Quantitative analysis of Munsell color data from archeological ceramics," *J. Archaeological Science: Rep.*, vol. 3, pp. 549–557, Sep. 2015.
- [2] F. Ganiyeva, "Porcelaine of eastern and western lifestyles: Similarities and differences," Vakanüvis-Uluslararası Tarih Araştırmaları Dergisi, vol. 8, no. 2, pp. 1513–1531, 2023.
- [3] D. Bleichmar and P. C. Mancall, Eds., Collecting Across Cultures: Material Exchanges in the Early Modern Atlantic World. Philadelphia, PA, USA: Univ. of Pennsylvania Press, 2011. [Online]. Available: http://www.jstor.org/stable/j.ctt3fhbdc
- [4] P. M. Rice, Pottery Analysis: A Sourcebook. Chicago, IL, USA: Univ. of Chicago Press, 1987, p. 592.
- [5] P. A. McAnany and S. L. L. Varela, "Re-creating the formative Maya village of k'axob: Chronology, ceramic complexes, and ancestors in architectural context," *Ancient Mesoamerica*, vol. 10, no. 1, pp. 147–168, Jan. 1999.
- [6] H. Wang, "A brief overview of ancient ceramic identification methods," *Identificat. Appreciation Cultual Relics*, vol. 15, no. 15, pp. 70–71, 2019. [Online]. Available: https://caod.oriprobe.com/articles/57134375/gu_tao_ ci_jian_ding_fang_fa_jian_shu_.htm
- [7] M. Lyons, "Ceramic fabric classification of petrographic thin sections with deep learning," *J. Comput. Appl. Archaeology*, vol. 4, no. 1, p. 188, Sep. 2021.
- [8] J. Wu, H. Luo, J. Li, H. Wang, X. Lu, and J. Wu, "Source and chronology analysis of ancient Chinese ceramic," *J. Chin. Ceram. Soc.*, vol. S1, no. S1, pp. 39–43, 2007. [Online]. Available: https://caod.oriprobe.com/ articles/28433671/PRODUCTION_SITES_AND_PERIODS_OF_ANCI ENT_CERAMICS.htm
- [9] Z. G. Portakal-Uçar, S. Akça-Özalp, M. Oglakci, A. Barad, and M. Topaksu, "Assessment of structural and thermoluminescence characteristics of porcelain used as an electrical insulator," *J. Amer. Ceram. Soc.*, vol. 107, no. 1, pp. 640–658, Jan. 2024.
- [10] A. Bezur and F. Casadio, "The analysis of porcelain using handheld and portable X-ray fluorescence spectrometers," in *Handheld XRF in Art and Archaeology*. A. N. Shugar and J. L. Mass, Eds. Leuven, Belgium: Univ. Press, 2012, pp. 249–311.
- [11] K. Gai, M. Qiu, H. Zhao, and X. Sun, "Resource management in sustainable cyber-physical systems using heterogeneous cloud computing," *IEEE Trans. Sustain. Comput.*, vol. 3, no. 2, pp. 60–72, Apr. 2018.
- [12] H. B. Abdalla, "A brief survey on big data: Technologies, terminologies and data-intensive applications," *J. Big Data*, vol. 9, no. 1, pp. 1–36, Nov. 2022.
- [13] L. Liu, G. Delnevo, and S. Mirri, "Unsupervised hyperspectral image segmentation of films: A hierarchical clustering-based approach," J. Big Data, vol. 10, no. 1, p. 31, Mar. 2023.
- [14] D. Teddy, J. Romain, C. Aladine, T. Sylvie, E. Matthieu, M. Lionel, and J. Sebastien, "Automatic pattern recognition on archaeological ceramic by 2D and 3D image analysis: A feasibility study," in *Proc. Int. Conf. Image Process. Theory, Tools Appl. (IPTA)*, Nov. 2015, pp. 224–228.
- [15] F. Stanco and A. M. Gueli, "Computer graphics solutions for pottery colors specification," in *Proc. SPIE*, vol. 8660, 2013, pp. 194–203.
- [16] M. Abadi, M. Khoudeir, and S. Marchand, "Gabor filter-based texture features to archaeological ceramic materials characterization," in *Image* and Signal Processing (Lecture Notes in Computer Science), vol. 7340, A. Elmoataz, D. Mammass, O. Lezoray, F. Nouboud, and D. Aboutajdine, Eds. Berlin, Germany: Springer, 2012, doi: 10.1007/978-3-642-31254-0_38.
- [17] K.-G. Wang, G.-H. Geng, and M.-Q. Zhou, "Texture-based ancient porcelain shards classifications using Gabor transform," in *Proc. 3rd Int. Conf. Intell. Syst. Knowl. Eng.*, Nov. 2008, pp. 699–702.

- [18] X. Zheng, H. Hu, and Z. Mingquan, "Porcelain shard images classification based on Gaussian color model," *J. Image Graph.*, vol. 17, no. 9, pp. 1115–1121, 2012.
- [19] E. Cumbajin, N. Rodrigues, P. Costa, R. Miragaia, L. Frazão, N. Costa, A. Fernández-Caballero, J. Carneiro, L. H. Buruberri, and A. Pereira, "A systematic review on deep learning with CNNs applied to surface defect detection," *J. Imag.*, vol. 9, no. 10, p. 193, Sep. 2023.
- [20] J. Zhang, X. Xiang, B. Xu, S. Huang, Y. Xiong, S. Ma, H. Fu, Y. Ma, H. Chen, and Z. Wu, "Rational design of high-entropy ceramics based on machine learning—A critical review," *Current Opinion Solid State Mater. Sci.*, vol. 27, no. 2, p. 101057, 2023.
- [21] L. Di Angelo, P. Di Stefano, and E. Guardiani, "A review of computerbased methods for classification and reconstruction of 3D high-density scanned archaeological pottery," *J. Cultural Heritage*, vol. 56, pp. 10–24, Jul. 2022.
- [22] Y. Ming, R. C. Me, J. K. Chen, and R. W. O. K. Rahmat, "A systematic review on virtual reality technology for ancient ceramic restoration," *Appl. Sci.*, vol. 13, no. 15, p. 8991, Aug. 2023.
- [23] A. Nightingale, "A guide to systematic literature reviews," Surgery (Oxford), vol. 27, no. 9, pp. 381–384, Sep. 2009.
- [24] D. Moher, "Preferred reporting items for systematic reviews and metaanalyses: The PRISMA statement," *Ann. Internal Med.*, vol. 151, no. 4, p. 264, Aug. 2009.
- [25] M. J. Page, J. E. McKenzie, P. M. Bossuyt, I. Boutron, T. C. Hoffmann, C. D. Mulrow, L. Shamseer, J. M. Tetzlaff, E. A. Akl, and S. E. Brennan, "The PRISMA 2020 statement: An updated guideline for reporting systematic reviews," *Int. J. surgery*, vol. 88, Mar. 2021, Art. no. 105906.
- [26] Y. Gao, "Ceramic texture extraction method based on convolution neural network and image processing," in *Proc. IEEE 4th Annu. Flagship India Council Int. Subsections Conf. (INDISCON)*, Aug. 2023, pp. 1–6.
- [27] L. Li, "Ceramic decoration extraction method based on computer vision and image processing," in *Proc. 2nd Int. Conf. Netw., Commun. Inf. Technol. (NetCIT)*, Dec. 2022, pp. 388–391.
- [28] Y. He, "The application of computer vision in the detection of ancient ceramic art," *Revista Argentina de Clinica Psicologica*, vol. 30, no. 2, p. 325, 2021.
- [29] Y. Yang, H. Wu, D. Yu, and C. Yang, "Ceramic type recognition algorithm based on ontology modeling and transfer learning," in *Proc. Int. Conf. Culture-Oriented Sci. Technol. (CoST)*, Aug. 2022, pp. 6–10.
- [30] Q. Liu, "Technological innovation in the recognition process of yaozhou kiln ware patterns based on image classification," *Soft Comput.*, 2023, doi: 10.1007/s00500-023-08528-8.
- [31] I. Mendonça, M. Miyaura, T. N. Fatyanosa, D. Yamaguchi, H. Sakai, H. Obata, and M. Aritsugi, "Classification of unexposed potsherd cavities by using deep learning," *J. Archaeological Sci., Rep.*, vol. 49, Jun. 2023, Art. no. 104003.
- [32] J. Sun, H. Lu, L. Qiao, X. Li, K. Chen, and W. Cao, "Identification of porcelain ewers in tang, song, and yuan dynasties by digital shape characterization," *Ceram. Int.*, vol. 49, no. 9, pp. 14246–14254, May 2023.
- [33] X. Yue, J. Chen, and Z. Zhuang, "Comparative study on image classification models of ancient ceramic types," in *Proc. IEEE 6th Int. Conf. Inf. Syst. Comput. Aided Educ. (ICISCAE)*, Sep. 2023, pp. 314–318.
- [34] L. Han, Y. Wang, and X. Wang, "Typical ceramic pattern extraction based on convolutional neural network," in *Proc. Int. Conf. Culture-Oriented Sci. Technol. (CoST)*, Oct. 2023, pp. 42–46.
- [35] C. Niu and M. Zhang, "Using image feature extraction to identification of ancient ceramics based on partial differential equation," *Adv. Math. Phys.*, vol. 2022, pp. 1–12, Jan. 2022.
- [36] Y. Liu, B. Liu, J. Yu, J. Xia, and C. Luo, "Automatic classification of blue and white porcelain sherds based on data augmentation and feature fusion," *Appl. Artif. Intell.*, vol. 36, no. 1, Dec. 2022, Art. no. 1994232.
- [37] L. Cardarelli, "A deep variational convolutional autoencoder for unsupervised features extraction of ceramic profiles. A case study from central Italy," *J. Archaeological Sci.*, vol. 144, Aug. 2022, Art. no. 105640.
- [38] J. Feng, Q. Zhou, R. Zhang, Y. Wang, and H. Luo, "Intelligent dating of Chinese ancient ceramics based on convolutional neural network," *J. Ceramics*, vol. 43, no. 1, pp. 145–152, 2022.
- [39] O. Chaowalit and P. Kuntitan, "Using deep learning for the image recognition of motifs on the center of sukhothai ceramics," *Current Appl. Sci. Technol.*, vol. 22, no. 2, Jul. 2021, Art. no. 55003.
- [40] L. M. Pawlowicz and C. E. Downum, "Applications of deep learning to decorated ceramic typology and classification: A case study using tusayan white ware from northeast Arizona," *J. Archaeological Sci.*, vol. 130, Jun. 2021, Art. no. 105375.

- [41] J. H. Yi, W. Kang, S.-E. Kim, D. Park, and J.-H. Hong, "Smart culture lens: An application that analyzes the visual elements of ceramics," *IEEE Access*, vol. 9, pp. 42868–42883, 2021.
- [42] J. Ma, Y. Peng, W. Cheng, M. Qiu, and Y. Nie, "Identification method of ancient ceramics revision," in *Proc. 8th IEEE Int. Conf. Cyber Secur. Cloud Comput. (CSCloud), 7th IEEE Int. Conf. Edge Comput. Scalable Cloud (EdgeCom)*, Jun. 2021, pp. 213–218.
- [43] K. Dia, V. L. Coli, J. Leblond, L. Gomart, D. Binder, and L. Blanc-Féraud, "Applications of learning methods to imaging issues in archaeology, regarding ancient ceramic manufacturing," in *Proc. 2nd Int. Conf. Deep Learn. Theory Appl.*, 2021, pp. 109–116.
- [44] J. Zhang, Q. Zhou, Y. Wang, and H. Luo, "Characteristics of banana leaf pattern for ceramics in Yuan, Ming and Qing Dynasties," *J. Chin. Ceram. Soc.*, vol. 48, no. 9, pp. 1367–1375, 2020.
- [45] T. Mu, F. Wang, X. Wang, and H. Luo, "Research on ancient ceramic identification by artificial intelligence," *Ceram. Int.*, vol. 45, no. 14, pp. 18140–18146, Oct. 2019.
- [46] J. Zhou, Y. Lu, K. Smith, C. Wilder, S. Wang, P. Sagona, and B. Torkian, "A framework for design identification on heritage objects," in *Proc. Pract. Exper. Adv. Res. Comput. Rise Mach. (learning)*, Jul. 2019, pp. 1–8.
- [47] I. Hein, A. Rojas-Domínguez, M. Ornelas, G. D'Ercole, and L. Peloschek, "Automated classification of archaeological ceramic materials by means of texture measures," *J. Archaeological Science: Rep.*, vol. 21, pp. 921–928, Oct. 2018.
- [48] S. H. Bickler, "Machine learning identification and classification of historic ceramics," *Archaeology New Zealand*, vol. 61, no. 1, pp. 21–33, 2018. Accessed: Jul. 27, 2022. [Online]. Available: https://www.researchgate. net/publication/323302055
- [49] O. Koval, "Using supervised machine learning to classify ceramic fabrics," M.A. thesis, Dept. Anthropol., Univ. Manitoba, 2018.
- [50] K. A. Maaroof, "Classification archaeological fragments into groups," *Res. J. Appl. Sci., Eng. Technol.*, vol. 14, no. 9, pp. 324–333, 2017.
- [51] P. Zhou and K. Wang, "Porcelain image classification based on semisupervised mean shift clustering," in *Proc. 8th IEEE Int. Conf. Softw. Eng. Service Sci. (ICSESS)*, Nov. 2017, pp. 791–797.
- [52] Z. Weng, Y. Guan, and H. Luo, "Machine vision based classification and identification for non-destructive authentication of ancient ceramic," *J. Chin. Ceram. Soc.*, vol. 45, no. 12, pp. 1833–1842, 2017. [Online]. Available: https://caod.oriprobe.com/articles/52650153/Machine_Vision_ Based_Classification_and_Identifica.htm
- [53] A. Rasheed and J. Nordin, "Using both HSV color and texture features to classify archaeological fragments," *Res. J. Appl. Sci., Eng. Technol.*, vol. 10, no. 12, pp. 1396–1403, Aug. 2015.
- [54] C. Piccoli, P. Aparajeya, G. T. Papadopoulos, J. Bintliff, F. F. Leymarie, P. Bes, M. van der Enden, J. Poblome, and P. Daras, "Towards the automatic classification of pottery sherds: Two complementary approaches," *Across Space Time*, vol. 463, no. 2, pp. 463–474, 2015.
- [55] K. Aryafar and P. L. Jeppson, "A semantic analysis approach to thin-shell ceramic fragment classification," *Int. J. Heritage Digit. Era*, vol. 2, no. 2, pp. 291–306, Jun. 2013.
- [56] M. Makridis and P. Daras, "Automatic classification of archaeological pottery sherds," *J. Comput. Cultural Heritage*, vol. 5, no. 4, pp. 1–21, Dec. 2012.
- [57] J. Y. Koh, D. Fried, and R. R. Salakhutdinov, "Generating images with multimodal language models," in *Proc. Adv. Neural Inf. Process. Syst.*, vol. 36, 2024, pp. 1–13.
- [58] T. S. Huang, W. F. Schreiber, and O. J. Tretiak, "Image processing," *Proc. IEEE*, vol. 59, no. 11, pp. 1586–1609, Nov. 1971.
- [59] W. K. Pratt, Digital Image Processing: PIKS Scientific Inside, vol. 4. Hoboken, NJ, USA: Wiley, 2007.
- [60] R. M. Haralick and L. G. Shapiro, "Image segmentation techniques," *Comput. Vis., Graph., Image Process.*, vol. 29, no. 1, pp. 100–132, 1985.
- [61] Z. Li, F. Liu, W. Yang, S. Peng, and J. Zhou, "A survey of convolutional neural networks: Analysis, applications, and prospects," *IEEE Trans. Neural Netw. Learn. Syst.*, vol. 33, no. 12, pp. 6999–7019, Dec. 2022.
- [62] Z. Zhang, Z. Gong, and Q. Hong, "A survey on: Application of transformer in computer vision," in *Proc. 8th Int. Conf. Intell. Syst. Image Process.*, 2021, pp. 21–28.
- [63] A. Vaswani, N. Shazeer, N. Parmar, J. Uszkoreit, L. Jones, A. N. Gomez, Ł. Kaiser, and I. Polosukhin, "Attention is all you need," in *Proc. Adv. Neural Inf. Process. Syst.*, vol. 30, 2017, pp. 1–11.
- [64] S. V. Mahadevkar, B. Khemani, S. Patil, K. Kotecha, D. R. Vora, A. Abraham, and L. A. Gabralla, "A review on machine learning styles in computer vision—Techniques and future directions," *IEEE Access*, vol. 10, pp. 107293–107329, 2022.



ZIYAO LING received the M.S. degree in science for the conservation and restoration of cultural heritage, in 2021. She is currently pursuing the Ph.D. degree in cultural and environmental heritage with the University of Bologna. Her Ph.D. research focused on applying computer vision techniques for the identification of archaeological ceramics.



PAOLA SALOMONI received the Italian Laurea degree (Hons.) in computer science from the University of Bologna, in October 1992.

From 1992 to 1995, she was a Research Staff of computer science degree course with the University of Bologna, Cesena. She was a Research Associate and an Associate Professor with the Department of Computer Science, from 1995 to 2001 and from 2001 to 2019, respectively. From 2015 to 2020, she was a Vice Rector of

digital technologies with the University of Bologna. She is currently a Full Professor of computer science with the Department of Computer Science and Engineering, University of Bologna. Her research interests include the following ones: design and implementation of distributed multimedia systems, design and implementation of e-learning environments, accessibility, content transcoding, and adaptation.



GIOVANNI DELNEVO (Member, IEEE) received the B.S. degree in computer science and information technology, the M.S. degree in computer science and engineering, and the Ph.D. degree in data science and computation from the University of Bologna, Bologna, Italy, in 2013, 2016, and 2022, respectively.

Since 2023, he has been a Junior Assistant Professor with the Department of Computer Science and Engineering, University of Bologna. His

research interests include human–AI interaction, machine learning, and human–computer interaction.



SILVIA MIRRI (Member, IEEE) received the M.S. and Ph.D. degrees in computer science from the University of Bologna, Italy, in 2002 and 2007, respectively.

From 2008 to 2020, she was a Research Assistant with the Department of Computer Science and Engineering, University of Bologna, where she is currently an Associate Professor. She authored more than 170 articles, including 38 articles in *Scientific Journals*. She has investigated HCI

topics related to specific contexts, such as digital humanities and cultural heritage, smart environments, and industry 4.0. Her research interests include human–computer interaction, ranging from content adaptation and transcoding for multimedia digital content to accessibility of digital technologies to meet users with disabilities needs.

. . .

Open Access funding provided by 'Alma Mater Studiorum - Università di Bologna' within the CRUI CARE Agreement